

# Stability analysis of systems with delay-dependent coefficients and commensurate delays

Chi Jin, Keqin Gu, Qian Ma, Silviu-Iulian Niculescu, Islam Boussaada

# ▶ To cite this version:

Chi Jin, Keqin Gu, Qian Ma, Silviu-Iulian Niculescu, Islam Boussaada. Stability analysis of systems with delay-dependent coefficients and commensurate delays. Mathematics of Control, Signals, and Systems, 2024, 10.1007/s00498-024-00399-0. hal-04710700

# HAL Id: hal-04710700 https://centralesupelec.hal.science/hal-04710700v1

Submitted on 26 Sep 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# Stability Analysis of Systems with Delay-Dependent Coefficients and Commensurate Delays

Chi Jin<sup>1</sup>, Keqin Gu<sup>2\*</sup>, Qian Ma<sup>3</sup>, Silviu-Iulian Niculescu<sup>4</sup>, Islam Boussaada<sup>4,5</sup>

 <sup>1</sup>TuSimple Inc., Room 1705, No. 489 Zhengli Road, Shanghai,China.
 <sup>2\*</sup>Department of Mechanical & Mechatronics Engineering, Southern Illinois University Edwardsville,Edwardsville, Illinois,62025, USA.
 <sup>3</sup>School of Automation, Nanjing University of Science and Technology, Nanjing 210094, China.
 <sup>4</sup>Université Paris-Saclay, CNRS, CentraleSupélec, Inria, Laboratoire des Signaux et Systèmes (L2S), F-91190, Gif-sur-Yvette, France.
 <sup>5</sup>Institut Polytechnique des Sciences Avancées (IPSA), 63 boulevard de

Brandebourg, 94200 Ivry-sur-Seine, France.

\*Corresponding author(s). E-mail(s): kgu@siue.edu; Contributing authors: jinchitrue@163.com; qianmashine@gmail.com; Silviu.Niculescu@l2s.centralesupelec.fr; islam.boussaada@l2s.centralesupelec.fr;

#### Abstract

This paper develops a method of stability analysis of linear time-delay systems with commensurate delays and delay-dependent coefficients. The method is based on a D-decomposition formulation that consists of identifying the critical pairs of delay and frequency, and determining the corresponding crossing directions. The process of identifying the critical pairs consists of a magnitude condition and a phase condition. The magnitude condition utilizes the Orlando's formula, and generates frequency curves within the delay interval of interest. Such frequency curves correspond to the the delay-frequency pairs such that the decomposition equation has at least one solution on the unit circle. The delay interval of interest is divided into continuous frequency curve intervals (CFCIs). Under some non-degeneracy assumptions, the number of frequency curves remains constant within each CFCI, and the associated decomposition equation has one and only

one solution on the unit circle at any point on a frequency curve. By traversing through the frequency curves, all the crossing points can be identified. The crossing direction is related to the sign of the lowest-order nonzero derivative of the phase angle with respect to the delay, which is a generalization of the existing literature even for the case with single delay. This conclusion allows one to determine the crossing direction by examining the phase angle vs delay diagram. An example is presented to illustrate how a stability analysis can be conducted if some non-degeneracy assumptions are violated.

**Keywords:** Stability, differential-difference equations, time-delay systems, delay-dependent coefficients, commensurate delays

## 1 Introduction

This paper develops a method of analyzing stability of linear time-delay systems described by differential-difference equations with commensurate delays and delay-dependent coefficients. The corresponding characteristic equation has the following general form

$$D(\lambda,\tau) = \sum_{k=0}^{M} P_k(\lambda,\tau) e^{-k\lambda\tau} = 0, \qquad (1)$$

where  $P_k(\lambda, \tau)$  for each given k is continuous with respect to the delay  $\tau$  and is a polynomial of the Laplace transform variable  $\lambda$  with real coefficients for any given  $\tau \in [\tau^l, \tau^u]$ . Such a system may arise from various scientific disciplines, such as population models with age structure [3], the stellar dynamos [40], and hematopoiesis dynamics [9]. The delay parameter in the coefficients may also arise from approximating derivative action by finite-difference in feedback control [23, 24, 36–38] (especially, see [29] for an example of commensurate delay approximation of derivative in feedback control), linearization of nonlinear models about equilibria that depend on the delay, and the problem of requiring guaranteed exponential convergent rate as will be shown in Section 2.

In spite of their practical importance, most existing approaches in the literature only consider systems with coefficients of the characteristic quasipolynominal independent of the delay. A notable exception is due to Berreta and Kuang [3], where an effective method was developed to determine stability of systems with a single delay and delay-dependent coefficients. In [20], this was generalized to a wider-class of systems, and the structure of the set of critical delay-frequency pairs are clearly described.

In [21], which may be considered as a preliminary version of this paper, the results of [20] are extended to the case of commensurate delays. In addition, the crossing direction analysis has been extended from the case that uses only the first-order derivative to the case where higher-order derivatives are needed, which is a generalization even for the single delay case. The analysis in [21] was based on some assumptions that turn out to be either overly restrictive or unnecessary, and some proofs are either omitted

or only sketched. No procedure is given for the calculations of some quantities used in the stability analysis.

In this paper, a more careful analysis allows us to show the implications of the four nondegeneracy assumptions. As a result, some assumptions in [21] are either relaxed or eliminated. The resulting process for stability analysis is simpler and more parallel to the single delay case. Complete proofs are provided and made more rigorous, and procedure of calculating the quantities needed for stability analysis are provided. Specifically, Assumptions II, III and IV in [21] are relaxed to Assumptions II, IV and III in this paper. In the analytical crossing direction analysis, Assumption IIIa is introduced, and its relationship with Assumption III, which parallels the single delay case, is presented. A new theorem (Theorem 1) has been introduced, which made Assumption V in [21] unnecessary. This also made the stability analysis process described Section 3 of this paper simpler and more parallel to the single delay case given in [20]. An important basis of establishing Theorem 1 is understanding the cases where no unit circle solution exists even though the Orlando magnitude equation is satisfied (see the discussions after Lemma 2 and Examples 1, 2 and 3). In the crossing direction analysis described in Section C of [21], only a sketch for the proof of the main result, Theorem 1, was provided, and the proof for Proposition 3 (which supports Theorem 1) was omitted. Furthermore, no procedure for the calculation of the quantities in Theorem 1 was provided. In this paper, the presentation of crossing direction analysis has been reorganized, the proofs are provided and made more rigorous. Four appendices have been introduced to provide background, cover technical details, and provide procedure of calculting quantifies necessary for crossing direction analysis. Additional observations are made to facilitate the analysis, including the possibility of making judgement of crossing direction by observing the crossing of  $r2\pi$  lines by the phase vs delay diagram (see Corollary 2), and invariance of two quantities used in the analysis (see Proposition 1).

In the following, we will provide a more extensive review of the background literature and highlight the new results in this paper and their significance.

### 1.1 *D*-decomposition

D-decomposition method [10] is a very effective method in the stability analysis of time-delay systems. The method is based on identifying the set of parameters, which we call stability crossing set [14], for which there is at least one characteristic root on the imaginary axis. In most practical cases, the characteristic roots depend on the parameters continuously. As a result, the stability crossing set divides the parameter space into regions with fixed number of characteristic roots on the right half-plane (RHP). By determining the direction the characteristic roots cross the imaginary axis, the number the RHP characteristic roots in each such region can be determined, especially, the parameter regions where there is no RHP characteristic root (thus the system is stable) can be identified.

When the parameter concerned is the delay, the *D*-decomposition method is also known as the  $\tau$ -decomposition method [27]. Due to periodic nature of the exponential function with imaginary exponent, significant complication may arise as compared with *D*-decomposition problem where no delay parameter is involved.

## 1.2 System with single delay and delay-dependent coefficients

In [3], the crossing direction in a neighborhood of a crossing pair was studied. In [20], an assumption in [3] was relaxed, and a method of searching all critical pairs was developed. In the following, we will outline the main idea of the method developed in [20] using a slightly different notation and terminology in order to make its relationship with the current paper more clear. For a system with characteristic equation

$$D_{(s)}(\lambda,\tau) = P(\lambda,\tau) + Q(\lambda,\tau)e^{-\lambda\tau} = 0,$$
(2)

using the same principle as used in [39] for the single delay case, the critical pairs  $(\omega, \tau)$  may be identified by using a magnitude condition

$$F_{(s)}(\omega,\tau) = P(j\omega,\tau)P(-j\omega,\tau) - Q(j\omega,\tau)Q(-j\omega,\tau) = 0$$
(3)

and a phase condition

$$\angle \theta_{(s)}(\omega,\tau) = \angle P(j\omega,\tau) - \angle Q(j\omega,\tau) + \omega\tau + \pi = 2r\pi.$$
(4)

Under nondegeneracy assumptions, the interval of interest  $\mathcal{T}$  may be divided into subintervals  $\mathcal{T}^{(i)} = [\tau^{(i-1)}, \tau^{(i)}]$ , which will be called Continuous Frequency Curve Intervals (CFCIs) in this paper. The dividing points  $\tau^{(i)}$ , i = 1, 2, ..., K - 1 satisfy

$$\partial_{\omega} F_{(s)}(\omega, \tau) = 0 \tag{5}$$

for some  $\omega$ . The number of real solutions of (3) for  $\omega$  remains constant within the interior of each subinterval  $\mathcal{T}^{(i)}$ , and these solutions are continuous functions  $\omega_k^{(i)}(\tau)$ ,  $k = 1, 2, \ldots, m^{(i)}$  of  $\tau$ . By traversing through these curves, all the critical pairs can be identified by using the phase condition (4).

## 1.3 System with multiple delays

Many practical systems have multiple delays. Such systems have a characteristic equation of the form

$$D(\lambda) = \sum_{k=0}^{M} P_k(\lambda) e^{-\tau_{(k)}\lambda} = 0,$$
(6)

where  $P_k(\lambda)$  are polynomials. Earlier studies consider systems with two delays. See [16, 32, 34] for the special case of  $P_k(\lambda) = \text{constant}$ , [35, 42] for two studies of the general case, [15] for a complete description of the problem structure and a procedure of stability analysis. When the coefficient polynomials  $P_k$  also depend on the delays, [1] extended the method in [15] to an interesting special class of such systems. Unfortunately, a thorough analysis of such systems with more than two delays becomes much more complicated [5, 11, 14].

## 1.4 System with commensurate delays

In some systems, the delays  $\tau_{(k)}$  in (6) are constrained to be commensurate, i.e.,  $\tau_{(k)} = k\tau$  for some  $\tau > 0$ . Such systems arise naturally in practice, such as from using more than two terms to more accurately approximate derivatives in feedback control [38] and from multivariable differential-difference equations with a single delay as will be shown in Section 2. The stability analysis for such systems with delay-independent coefficients has been carried out in, for example, [6, 7, 18, 19, 39].

This paper considers the stability analysis of systems with commensurate delays and delay-dependent coefficients. Similar to the single delay case discussed in [20], the conditions for the critical pairs consist of a magnitude condition and a phase condition. However, the magnitude condition is defined implicitly by requiring the following equation

$$\sum_{k=0}^{M} P_k(j\omega,\tau) x^k = 0 \tag{7}$$

to have at least one solution on the unit circle. The phase condition is expressed in terms of the phase angles of such unit circle solutions.

A method to write the magnitude condition in an explicit form is to apply the Orlando's formula [22, 33], which is presented in Appendix A. However, a number of complications arise: 1) There may be cases that such explicit form of magnitude condition is satisfied, but the equation (7) does not have any solution on the unit circle; 2) The equation (7) may have more than one solutions on the unit circle. To circumvent these complications, some assumptions are made regarding the invariance of unit circle solutions of (7) in [21], which may be considered as a preliminary version of this paper. In this paper, it was shown that the nondegeneracy assumptions imply that the explicit form of magnitude condition guarantees that the equation (7) has one and only one solution on the unit circle when the delay is restricted to the interior of any CFCI. This significantly simplified the structure of the problem. As a result, it is possible to make the solution process more parallel to the single delay case.

In addition, this paper also generalized the analytical crossing condition from the case that can be determined by the first-order derivative to the one that requires higher-order derivatives.

Arriving at these results depends on an analysis of the derivatives of polynomial roots with respect to parameters which may be constrained by another equation. Such analysis is more subtle, and requires some additional tools as compared with [20]. For example, when a multiple solutions of polynomial is concerned, a puiseux series [25, 26] is needed for such analysis, and a higher-order derivatives of one variable with respect to one or more other variables defined by implicit function relations are needed.

### 1.5 Outline of the paper

Section 2 sets up the problem, defines the magnitude and phase conditions, introduces the nondegeneracy assumptions, and discusses the implications of these assumptions. Section 3 discusses the stability analysis, including the description of continuous frequency curves and continuous frequency curve intervals, the determination of critical pairs, the crossing directions, culminating in the determination of stable delay intervals. Section 4 presents an analytical method to determine the crossing direction of critial pairs that generalizes the existing results to one that may require higher-order derivatives, and relate the crossing direction of the characteristic roots at the imaginary axis to that of the phase angle as a function of delay at some designated horizontal lines. Some invariant results are also presented. Section 5 presents some illustrative examples, including one that illustrates the stability analysis of a system that violates some nondegeneracy assumptions. Section 6 presents some conclusions of this paper and potential further research. Four appendices are included to provide supplementary materials. Appendix A presents the Orlando's formula. Appendix B discusses the derivatives of one variable with respect to one or more other variables defined by implicit functions. Appendix C discusses the derivatives of phase angle as a function of delay. Appendix D presents the proof of one of the main theorems (Theorem 2).

## 1.6 Notation

For a complex number c,  $\mathcal{R}(c)$ ,  $\mathcal{I}(c)$  and  $\overline{c}$  denote its real part, imaginary part and complex conjugate, respectively. The unit circle in the complex plane is denoted by S.  $\mathbb{R}$  stands for the set of real numbers,  $\mathbb{R}_+$  for positive reals, and  $\overline{\mathbb{R}}_+$  for nonnegative reals. For a function defined in a closed interval, say [a, b], its derivatives at a and bare understood as the right and left derivative, respectively. We sometimes use  $\partial$  with a subscript to denote partial derivatives. For instance,  $\partial_{\lambda}D(\lambda,\tau) := \frac{\partial D(\lambda,\tau)}{\partial \lambda}$ . For a function with two variables, we may write one variable as a subscript to indicate that we consider this variable as a fixed parameter. For example, if  $D(\lambda,\tau)$  is continuous function of  $\tau$  and polynomial of  $\lambda$ , then  $D_{\tau}(\lambda) = D(\lambda,\tau)$ . With this notation, we can say  $D_{\tau}(\lambda)$  is a polynomial. Similarly, for a function of three variables such as  $\Phi(\omega,\tau,x)$ , then  $\Phi_{\omega\tau}(x) = \Phi(\omega,\tau,x)$ , but we consider  $\Phi_{\omega\tau}(x)$  as a function of x, and  $\omega$  and  $\tau$  as fixed parameters. For a function of single variable f(x), its derivative is denoted by f'(x). For a polynomial  $P(\lambda)$ , its degree is denoted as deg $(P)^1$ .

## 2 Systems with commensurate delays

Consider a system represented by a multivariable differential-difference equation with the characteristic equation given in the equation (1), where each  $P_k(\lambda, \tau)$  is continuous with respect to the delay  $\tau$  and is a polynomial of the Laplace transform variable  $\lambda$  with real coefficients for any given  $\tau \in \mathcal{T}$ , and  $\mathcal{T} = [\tau^l, \tau^u]$  is the delay interval of interest. For a  $\tau \in \mathcal{T}$ , let  $N^u(\tau)$  be the number of characteristic roots of (1) on the strict RHP for the given  $\tau$ . We assume  $N^u(\tau^l)$  is known (well-known methods such as the one given in [39] may be used to determine this). Our objective is to find the set of  $\tau$  contained in  $\mathcal{T}$  for which the system with characteristic equation (1) is asymptotically stable.

The characteristic equation given in (1) has commensurate delays and delaydependent coefficients. As indicated in the introduction, this form of characteristic equation appears frequently in practical systems. As a specific example, consider a

<sup>&</sup>lt;sup>1</sup>This was denoted as " $\operatorname{ord}(P)$ " representing the order of the polynomial in [20, 21]. We now use the term "degree" instead to avoid potential ambiguity.

system represented by the following multi-variable differential-difference equation,

$$\dot{x}(t) = A_1(\tau)x(t) + A_2(\tau)x(t-\tau),$$

where  $A_1$  and  $A_2$  are real matrices that depend on  $\tau$ . It is easily verified that its characteristic equation is indeed of the form (1) with M equal to the dimension of x. Note that in this case commensurate delays may appear in the corresponding characteristic equation even though there is only one delay in the original differential-difference equation. In addition, characteristic equations with delay-dependent coefficients may arise from the  $\alpha$ -stability analysis of systems with delay-free coefficients. Consider a system with the characteristic equation [6], [18]:

$$\lambda + e^{-\tau\lambda} + e^{-2\tau\lambda} = 0. \tag{8}$$

The system is said to be  $\alpha$ -stable if the real parts of all the roots of (8) are less than  $-\alpha$ . Replacing  $\lambda$  with  $\lambda - \alpha$  in (8), we obtain

$$\lambda - \alpha + e^{\alpha \tau} e^{-\tau \lambda} + e^{2\alpha \tau} e^{-2\tau \lambda} = 0.$$
<sup>(9)</sup>

It is easy to see that the  $\alpha$ -stability of the system with characteristic equation (8) is equivalent to the asymptotic stability of the system with characteristic equation (9), which has delay-dependent coefficients, even though the coefficients of (8) are independent of the delay  $\tau$ .

It is instrumental to define

$$\Phi(\omega,\tau,x) = \sum_{k=0}^{M} P_k(j\omega,\tau)x^k,$$
(10)

and

$$\Psi(\omega,\tau,x) = \sum_{k=0}^{M} P_k(-j\omega,\tau) x^{M-k}.$$
(11)

According to the definition,

$$D(j\omega,\tau) = \Phi(\omega,\tau,e^{-j\omega\tau}).$$
(12)

Then the following lemma is obvious.

**Lemma 1.** The characteristic equation (1) has a solution for  $\lambda$  on the imaginary axis, i.e.,

$$D_{\tau}(j\omega) = 0 \tag{13}$$

for some  $\omega \in \mathbb{R}$  if and only if

$$\Phi_{\omega\tau}(x) = 0, \tag{14}$$

or equivalently,

$$\Psi_{\omega\tau}(x) = 0, \tag{15}$$

has a solution on the unit circle  $x \in \mathbb{S}$  such that

$$\angle x = -\omega\tau + r2\pi \tag{16}$$

#### for some integer r.

*Proof.* The equations (14) with  $x \in S$  and (16) are equivalent to (13) in view of (12). Therefore, it is sufficient to show that (14) and (15) are equivalent for  $x \in S$ . Take complex conjugate of (14), multiply the resulting equation by  $x^M$  and observe that  $\bar{x} = x^{-1}$  when  $x \in S$ , we obtain (15). This process is reversible.

A crucial step in carrying out the  $\tau$ -decomposition process is to identify all the pairs  $(\omega, \tau) \in \mathbb{R}_+ \times \mathcal{T}$  such that (13) is satisfied. Parallel to the single delay case discussed in [20], Lemma 1 decomposes the requirement (13) into two parts: the *magnitude condition* (i.e., (14) is satisfied for some  $x \in \mathbb{S}$ ) and the *phase condition* (16). Because of its role in the decomposition to the magnitude condition and the phase condition, the equation (14) will be referred to as the *decomposition equation*. Similarly, the equation (15) will be referred to as the *alternate decomposition equation*.

While the magnitude condition in the single delay case is very simple, the commensurate delay counterpart discussed here is defined implicitly. Fortunately, an explicit condition is still possible by utilizing the Orlando's formula, which is restated in Appendix A. Define the Schur-Cohn-Fujiwara matrix [6, 7, 18, 19],

$$\Theta(\omega,\tau) = \Psi^H(\omega,\tau,S)\Psi(\omega,\tau,S) - \Phi^H(\omega,\tau,S)\Phi(\omega,\tau,S),$$
(17)

where a supercript H represents conjugate transpose of the matrix, and S is the  $M \times M$  shift matrix defined in (A4) in Appendix A. Define

$$F(\omega, \tau) = -\det \Theta(\omega, \tau). \tag{18}$$

Then the magnitude condition can be expressed in terms of F in view of the following lemma.

**Lemma 2.** If  $\omega$  and  $\tau$  are such that

$$P_M(j\omega,\tau) \neq 0,\tag{19}$$

then

$$F(\omega,\tau) = -|P_M(j\omega,\tau)|^{2M} \prod_{i=1}^{M} \prod_{k=1}^{M} (1 - x_{\Phi i} \bar{x}_{\Phi k}),$$
(20)

where  $x_{\phi i}, i = 1, 2, ..., M$  are the M solutions of the decomposition equation (14). If  $\omega$  and  $\tau$  satisfy

$$P_0(j\omega,\tau) \neq 0,\tag{21}$$

then

$$F(\omega,\tau) = (-1)^{M+1} \left| P_0(j\omega,\tau) \right|^{2M} \prod_{i=1}^M \prod_{k=1}^M (1 - x_{\Psi i} \bar{x}_{\Psi k}), \tag{22}$$

where  $x_{\Psi i}$ , i = 1, 2, ..., M are the M solutions of the alternate decomposition equation (15).

*Proof.* The expression (20) is obtained by applying the Orlando's formula given in Appendix A to the polynomial  $\Phi_{\omega\tau}(x)$ . The expression (22) is obtained by applying the Orlando's formula to the polynomial  $\Psi_{\omega\tau}(x)$ .

It is immediately clear from Lemma 2 that

$$F(\omega, \tau) = 0 \tag{23}$$

whenever the decomposition equation (14) admits a solution on the unit circle unless both  $P_M(j\omega,\tau)$  and  $P_0(j\omega,\tau)$  vanish. We will refer to (23) as the Orlando magnitude equation.

Suppose that either (19) or (21) is satisfied. It is still possible for the Orlando magnitude equation (23) to be satisfied even if the decomposition equation (14) has no solution on the unit circle. This case arises when there are two solutions  $x_i$  and  $x_k$  of (14) (when  $P_M(\omega, \tau) \neq 0$ ) or (15) (when  $P_0(\omega, \tau) \neq 0$ ) that are on the same radial line and their magnitudes are reciprocal of each other:

$$|x_i| = 1/|x_k|, (24)$$

$$\angle x_i = \angle x_k + r2\pi, \quad r \text{ integer.} \tag{25}$$

The above can be easily seen from (20) and (22).

We will introduce some assumptions in order to avoid some degenerate cases and present our approach within a limited amount of space. The cases that violate one or more assumptions can usually be treated individually based on the same principle presented here. An example will be presented in Section 5.

Assumption I. For all  $\tau \in \mathcal{T}$ ,

$$\deg(P_{0\tau}) = n, \tag{26}$$

$$\deg(P_{k\tau}) \le n, \ k = 1, 2, \cdots, M.$$

$$(27)$$

Furthermore, all the roots of the following equation

$$s^{M} + c_{1\tau}s^{M-1} + c_{2\tau}s^{M-2} + \dots + c_{M-1,\tau}s + c_{M} = 0$$
(28)

satisfy |s| < 1, where

$$c_{k\tau} = \lim_{\lambda \to \infty} \frac{P_{k\tau}(\lambda)}{P_{0\tau}(\lambda)}, \ k = 1, 2, \dots, M$$

		2	
3			

Assumption II. No  $(\omega, \tau) \in \mathbb{R} \times \mathcal{T}$  simultaneously satisfies

$$P_M(j\omega,\tau) = 0, (29)$$

and

$$P_0(j\omega,\tau) = 0. \tag{30}$$

Assumption III. Any  $(\omega^*, \tau^*) \in in \mathbb{R}_+ \times \mathcal{T}$  that satisfies (13) must also satisfy

$$\partial_{\lambda} D(\lambda, \tau)|_{\lambda = j\omega^*, \tau = \tau^*} \neq 0.$$
(31)

Furthermore, let  $\lambda(\tau)$  be the unique function implicitly defined by (1) in a sufficiently small neighborhood of  $\tau^*$  taking value within a small neighborhood of  $j\omega^*$ , then

$$\mathcal{R}(\lambda(\tau)) \neq 0 \quad \text{for all } \tau \neq \tau^*$$
 (32)

in this small neighborhood.

Assumption IV. There are only a finite number of  $(\omega, \tau) \in \mathbb{R}_+ \times \mathcal{T}$  that simultaneously satisfy the Orlando magnitude equation (23) and the following equation

$$\partial_{\omega} F(\omega, \tau) = 0. \tag{33}$$

In the remaining part of this paper, these four nondegeneracy assumptions will be assumed to be satisfied unless specifically pointed out.

The above four assumptions parallel those for the single delay case given in [20]. Indeed, Assumption I assures that  $\tau$ -decomposition is applicable. In this assumption, (28) is the difference equation associated with the system (1). When the system is of retarded type [12, 17], (27) is satisfied with strict inequality because  $c_{k\tau} = 0$ , k = $1, 2, \ldots M$ , and the assumption is automatically satisfied. When the system is of neutral type [12, 17], the assumption implies that there exists an  $\varepsilon > 0$  such that only a finite number of characteristic roots of the system (1) are on the right hand side of the vertical line  $\mathcal{R}(\lambda) = -\varepsilon$ , and they are continuous functions of  $\tau$  [12]. It should be pointed out that the assumption  $P_k(\lambda, \tau)$ ,  $k = 0, 1, \ldots, M$  be continuous with respect to  $\tau$  is essential. Indeed, [28] gives an example that a characteristic root may suddenly appear on the positive real axis when the characteristic quasipolynomial depends on  $\frac{1}{\tau}$  (and thus discontinuous at  $\tau = 0$ ).

Assumption II implies that the Orlando magnitude equation (23) is satisfied for all the pairs  $(\omega, \tau)$  such that the decomposition equation (14) has at least one root on the unit circle. Violation of this assumption would require two real variables  $\omega$  and  $\tau$  to satisfy two complex equations (29) and (30), which are equivalent to four real equations. Therefore, it is indeed rare that this assumption would be violated.

Assumption III means that all imaginary roots of  $D_{\tau}(\lambda)$  are simple, and the curve  $\lambda(\tau)$  intersects with the imaginary axis only at one point  $\lambda(\tau^*)$  in a small neighborhood of  $\tau^*$ . When  $D(\lambda, \tau)$  is differentiable with respect to  $\tau$ , then a sufficient condition for

the assumption is

$$\mathcal{R}(\lambda'(\tau^*)) \neq 0. \tag{34}$$

Assumption IV is a very minor restriction. Similar to the single delay case,  $F(\omega, \tau)$  is an even polynomial function of  $\omega$  for any given  $\tau$ . If the pair  $(\omega, \tau)$  satisfies both (23) and (33), then  $\omega$  is a multiple root of this polynomial.

The following theorem plays an important role in simplifying stability analysis.

**Theorem 1.** Let  $(\omega, \tau) \in \mathbb{R} \times \mathcal{T}$  satisfy the Orlando magnitude equation (23) and the following equation

$$\partial_{\omega} F(\omega, \tau) \neq 0. \tag{35}$$

Then, there is one and only one root of  $\Phi_{\omega\tau}(x)$  on the unit circle, and this root is simple.

*Proof.* Let  $(\omega^*, \tau) \in \mathbb{R} \times \mathcal{T}$  satisfy (23) and assume  $P_M(j\omega^*, \tau) \neq 0$ . It is sufficient to show that (35) must be violated if  $\Phi_{\omega^*\tau}(x)$  has either one of the following two types of roots:

I) At least two roots are on the unit circle;

II) Two roots  $x_i, x_k$  that are not on the unit circle, and satisfy (24) and (25).

We will first show Case I). This case can be further divided to two subcases:

Ia) At least two simple roots are on the unit circle;

Ib) There is a multiple root on the unit circle.

For Case Ia), according to the equation (20) in Lemma 2, we may write

$$F(\omega,\tau) = R(\omega,\tau)(1-x_i\bar{x}_i)(1-x_k\bar{x}_k), \qquad (36)$$

where  $R(\omega, \tau)$  is continuous, and  $x_i$  and  $x_k$  are continuously differentiable with respect to  $\omega$  at  $\omega^*$ . Furthermore,  $x_i^* = x_i(\omega^*), x_k^* = x_k(\omega^*)$ 

satisfy

$$x_i^* \bar{x}_i^* = x_k^* \bar{x}_k^* = 1. ag{37}$$

Apply Lemma 4 in Appendix B, we conclude that F is differentiable with respect to  $\omega$  at  $\omega^*$ , and

$$\partial_{\omega} F(\omega^*, \tau) = R(\omega^*, \tau) \frac{d}{d\omega} [(1 - x_i \bar{x}_i)(1 - x_k \bar{x}_k)]_{x_i = x_i^*, x_k = x_k^*} \\ = R(\omega^*, \tau) \{ (1 - x_i^* \bar{x}_i^*) [-x'_k(\omega^*) \bar{x}_k(\omega^*) - x_k(\omega^*) \bar{x}'_k(\omega^*)] \\ + [-x'_i(\omega^*) \bar{x}_i(\omega^*) - x_i(\omega^*) \bar{x}'_i(\omega^*)] (1 - x_k^* \bar{x}_k^*) \} \\ = 0$$

in view of (37).

For Case Ib), let the root  $x = x_0 \in S$  have multiplicity p when  $\omega = \omega^*$ . Then, for  $\omega = \omega^* + \varepsilon$ , the multiple root splits into p roots, and, from the equation (20) in Lemma 2,

$$F(\omega,\tau) = S(\omega,\tau) \prod_{i=1}^{p} \prod_{k=1}^{p} (1 - x_i \bar{x}_k),$$
(38)

where  $S(\omega, \tau)$  and  $x_i, i = 1, 2, \dots p$  are continuous. In general, these  $x_i$ 's are not differentiable with respect to  $\omega$  at  $\omega^*$ . Indeed[25, 26], they can be expressed using pulseux series as

$$x_k = x_0 + c_k \varepsilon^{1/p} + o(\varepsilon^{1/p}), \quad k = 1, 2, \dots p,$$
(39)

where  $c_k, k = 1, 2, ..., p$  are complex constants. It can be calculated that,

$$1 - x_i \bar{x}_k = 1 - (x_0 + c_i \varepsilon^{1/p} + o(\varepsilon^{1/p}))(x_0 + c_k \varepsilon^{1/p} + o(\varepsilon^{1/p}))$$
  
=  $1 - x_0 \bar{x}_0 - (x_0 \bar{c}_k + \bar{x}_0 c_i)\varepsilon^{1/p} + o(\varepsilon^{1/p})$   
=  $-(x_0 \bar{c}_k + \bar{x}_0 c_i)\varepsilon^{1/p} + o(\varepsilon^{1/p}).$ 

Therefore,

$$F(\omega,\tau) = S(\omega,\tau) \prod_{i=1}^{p} \prod_{k=1}^{p} (1-x_i \bar{x}_k)$$
$$= S(\omega,\tau) (c_{\Pi} \varepsilon^p + o(\varepsilon^p)),$$

where

$$c_{\Pi} = \prod_{i=1}^{p} \prod_{k=1}^{p} (-x_0 \bar{c}_k - \bar{x}_0 c_i).$$
(40)

As a result,

$$\partial_{\omega}F(\omega,\tau) = \lim_{\varepsilon \to 0} \frac{S(\omega,\tau)(c_{\Pi}\varepsilon^p + o(\varepsilon^p))}{\varepsilon} = 0.$$

For case II), we will only show the case where both  $x_i$  and  $x_k$  are simple. The case with multiple roots can be shown in a way similar to Case Ib). Similar to the case Ia), we may again write, from the equation (20) in Lemma 2,

$$F(\omega,\tau) = Q(\omega,\tau)(1-x_i\bar{x}_k)(1-x_k\bar{x}_i), \qquad (41)$$

where  $Q(\omega, \tau)$  is continuous, and  $x_i$  and  $x_k$  are differentiable at  $\omega = \omega^*$ , and  $x_i(\omega^*) = x_i^*, x_k(\omega^*) = x_k^*$  satisfy

$$x_i^* \bar{x}_k^* = x_k^* \bar{x}_i^* = 1. (42)$$

Apply Lemma 4 in Appendix B, we again conclude that F is differentiable with respect to  $\omega$  at  $\omega^*,$  and

$$\partial_{\omega} F(\omega^*, \tau) = Q(\omega^*, \tau) \{ (1 - x_i^* \bar{x}_k^*) [-x_k'(\omega^*) \bar{x}_i(\omega^*) - x_k(\omega^*) \bar{x}_i'(\omega^*)] \\ + [-x_i'(\omega^*) \bar{x}_k(\omega^*) - x_i(\omega^*) \bar{x}_k'(\omega^*)] (1 - x_k^* \bar{x}_i^*) \} \\ = 0.$$

Finally, if  $P_M(j\omega,\tau) = 0$ , then  $P_0(j\omega,\tau) \neq 0$  per Assumption II. The proof proceeds similarly using the expression (22) instead of (20).

In the following examples, the equation (23) is satisfied, but the condition (35) in Theorem 1 is violated for the given  $\omega$  and  $\tau$ . According to the theorem above,  $\Phi_{\omega\tau}(x)$  cannot have one and only one simple root on the unit circle. They illustrate a number of possible scenarios.

**Example 1.** Consider the characteristic equation (1) with

$$P_0 = -0.6\tau + 0.8\lambda, P_1 = -0.9\tau - 1.8\lambda, P_2 = 1.$$

At  $\omega = 1$  and  $\tau = 1$ ,  $\Phi_{\omega\tau}(x)$  has no root on the unit circle. However, it has two roots  $x_1 = 0.4 + 0.8j$  and  $x_2 = 0.5 + j$  satisfying (24) and (25). Example 2. Consider the characteristic equation (1) with

$$P_0 = -2\tau, P_1 = 1 - \lambda\tau, P_2 = \lambda\tau.$$

At  $\omega = 2$  and  $\tau = 0.5$ ,  $\Phi_{\omega\tau}(x)$  has two distinct simple roots  $x_1 = 1$  and  $x_2 = j$  on the unit circle.

**Example 3.** Consider the characteristic equation (1) with

$$P_0 = \lambda \tau, \ P_1 = -\sqrt{2} - \sqrt{2}\lambda \tau, \ P_2 = 0.64\tau^2.$$

At  $\omega = 0.8$  and  $\tau = 1.25$ ,  $\Phi_{\omega\tau}(x)$  has a double root  $x_1 = x_2 = \frac{\sqrt{2}}{2} + \frac{\sqrt{2}}{2}j$  on the unit circle.

## 3 Stability analysis

Due to Theorem 1, the general process of stability analysis for systems with commensurate delay can be made parallel to the one for the single delay case. The first step of the process is to determine the critical delay corresponding to the imaginary characteristic roots. These critical delays divide the delay interval of interest  $\mathcal{T}$  into subintervals. The number of characteristic roots on the RHP is fixed for  $\tau$  to be within each such subinterval. The second step is to determine the increment of RHP roots at each critical delay. This will allow us to determine the number of RHP roots for each delay subinterval, especially, those delay subintervals without RHP roots can be identified, and the stability analysis is complete.

#### 3.1 Determination of critical delays

The first step of our analysis is to identify all  $\tau$  values such that the characteristic equation has at least one imaginary root.

Recall that if  $\lambda = j\omega$  is an imaginary root of  $D_{\tau}(\lambda)$ ,  $\omega$  must be a real root of  $F_{\tau}(\omega)$ . Therefore it is natural to expect that the real roots of  $F_{\tau}(\omega)$  will play an important role in the analysis. We first show that  $\mathcal{T}$  can be decomposed into disjoint intervals in which the number of real roots of  $F_{\tau}(\omega)$  is invariant.

Let  $\tau^{(i)} \in \mathcal{T}, i = 1, 2, ..., K - 1$  be the set of  $\tau \in \mathcal{T}$  such that the equations (23) and (33) are simultaneously satisfied by such an  $\tau$  and some  $\omega \geq 0$  (as  $F_{\tau}(\omega)$  is an

even function of  $\omega$ , it is sufficient to restrict ourselves to studying  $\omega \geq 0$ ). Arrange this set in ascending order,

$$\tau^{(1)} < \tau^{(2)} < \dots < \tau^{(K-1)},$$

and write  $\tau^{(0)} = \tau^l, \tau^{(K)} = \tau^u$ . Then, the interval  $\mathcal{T}$  can be divided into K subintervals

$$\mathcal{T}^{(i)} = [\tau^{(i-1)}, \tau^{(i)}], \ i = 1, \dots, K.$$
(43)

Each subinterval  $\mathcal{T}^{(i)}$  is known as a continuous frequency curve interval (CFCI). It has been shown in Proposition 1 of [20] that, for all  $\tau$  within  $\mathcal{T}_o^{(i)}$ , the interior of the CFCI  $\mathcal{T}^{(i)}$ ,  $F_{\tau}(\omega)$  admits a fixed number of simple real roots. These roots are continuous functions of  $\tau$ , denoted as  $\omega_k^{(i)}(\tau)$ ,  $k = 1, 2, ..., m^{(i)}$ , and are known as frequency curves. The definition of these continuous functions (and the frequency curves) is extended to  $\mathcal{T}^{(i)}$  by requiring them to be continuous at  $\tau^{(i-1)}$  and  $\tau^i$ .

For any  $\tau \in \mathcal{T}_o^{(i)}$ , Theorem 1 implies that the decomposition equation

$$\Phi_{\omega_{k}^{(i)}(\tau),\tau}(x) = 0 \tag{44}$$

has one and only one solution on the unit circle. This has two important implications: First, the pair  $(\omega_k^{(i)}(\tau), \tau)$  satisfies the magnitude condition. Second, there is no need to consider more than one solutions of the decomposition equation (44) on the unit circle.

Let the unique unit circle solution of the decomposition equation (44) be denoted as  $z_k^{(i)}(\tau)$ , then  $z_k^{(i)}(\tau)$  is a continuous function of  $\tau$ , and we can extend its definition to the entire CFCI by requiring the function  $z_k^{(i)}(\tau)$  to be continuous at the two ends of  $\mathcal{T}^{(i)}$ .

Define

$$\theta_k^{(i)}(\tau) = \angle z_k^{(i)}(\tau) + \omega_k^{(i)}(\tau)\tau, \tag{45}$$

where  $\angle z_k^{(i)}(\tau)$  is the phase angel of the complex number  $z_k^{(i)}(\tau)$  and is required to be a continuous function of  $\tau$  in  $\mathcal{T}^{(i)}$ . Notice, the value of  $\angle z_k^{(i)}(\tau)$  is not necessarily restricted to any  $2\pi$  interval due to continuity requirement.

Using Lemma 1, we obtain the following result immediately.

**Corollary 1.** For any given *i*, a pair  $(\omega^*, \tau^*) \in \overline{\mathbb{R}}_+ \times \mathcal{T}^{(i)}$  satisfies (13) if and only if there exist some index *k* such that  $\omega^* = \omega_k^{(i)}(\tau^*)$  and

$$\theta_k^{(i)}(\tau^*) = 2r\pi \tag{46}$$

for some integer r.

Going through each interval  $\mathcal{T}^{(i)}$  and traversing through each curve  $\omega_k^{(i)}(\tau)$ ,  $k = 1, 2, \ldots, m^{(i)}$ , within  $\mathcal{T}^{(i)}$ , we may identify all  $\tau^* = \tau_l$ ,  $l = 1, 2, \ldots, L$  such that (46) holds for some integer r and indices k and i. Each such  $\tau_l$  is known as a critical

delay. For each given critical delay  $\tau_l$ , it is possible that (46) is satisfied for more than one k. Denote these k's as  $k_h, h = 1, 2, ..., H_l$ . Then we denote the corresponding  $\omega_{k_h}^{(i)}(\tau_l) \ge 0$  as  $\omega_{lh}, h = 1, 2, ..., H_l$ . In this way we can identify all the pairs  $(\omega_{lh}, \tau_l),$  $h = 1, 2, ..., H_l; l = 1, 2, ..., L$ , that satisfy (13).

#### 3.2 Determination of stable delay intervals

We now describe the second step of stability analysis mentioned at the beginning of the section. Under Assumption III, Equation (1) defines an implicit function  $\lambda(\tau)$  in a small neighborhood of  $\tau_l$ , and  $\lambda(\tau)$  lies in a small neighborhood of  $j\omega_{lh}$ , and satisfies (32) for  $\tau \neq \tau_l$ .

If  $\tau_l \neq \tau^l$ , define

$$\operatorname{Inc}(\omega_{lh}, \tau_l) = \lim_{\epsilon \to 0^+} \frac{\operatorname{sgn}\left(\mathcal{R}\left(\lambda(\tau_l + \epsilon)\right)\right) - \operatorname{sgn}\left(\mathcal{R}\left(\lambda(\tau_l - \epsilon)\right)\right)}{2}.$$
(47)

Due to Assumption III, possible values of  $\operatorname{Inc}(\omega_{lh}, \tau_l)$  are -1, 0, and 1. When  $\operatorname{Inc}(\omega_{lh}, \tau_l) = 1$ , a characteristic root moves from left half-plane (LHP) to RHP through  $j\omega_{lh}$  on the imaginary axis. The movement is in the opposite direction if  $\operatorname{Inc}(\omega_{lh}, \tau_l) = -1$ . The root touches the imaginary axis and returns to the same side of the imaginary axis without crossing it if  $\operatorname{Inc}(\omega_{lh}, \tau_l) = 0$ .

If  $\tau_l = \tau^l$ , which implies  $\tau_l = \tau_1$ , define instead

$$\operatorname{Inc}(\omega_{1h},\tau_1) = \max\{0, \lim_{\epsilon \to 0^+} \operatorname{sgn}\left(\mathcal{R}\left(\lambda(\tau_1 + \epsilon)\right)\right)\}.$$
(48)

Now let the number of RHP roots of  $D_{\tau}(\lambda)$  be  $N^{u}(\tau)$ . For any  $\tau \in \mathcal{T}, \tau \neq \tau_{l}$ ,  $l = 1, 2, \ldots, L$ , it is easy to see that the following relation holds:

$$N^{u}(\tau) = N^{u}(\tau^{l}) + \sum_{l=1}^{L_{\tau}} \operatorname{Inc}(\tau_{l}),$$
(49)

where  $L_{\tau} = \max\{l \mid \tau_l < \tau\}$  and

$$\operatorname{Inc}(\tau_l) = 2\sum_{h=1}^{H_l} \operatorname{Inc}(\omega_{lh}, \tau_l).$$
(50)

The coefficient 2 in the above equation is due to the fact that the crossing direction is the same at  $j\omega_{lh}$  and  $-j\omega_{lh}$ . The subintervals of  $\mathcal{T}$  corresponding to  $N^u(\tau) = 0$  can be determined. These subintervals of  $\mathcal{T}$  give the set of  $\tau$  corresponding to stable system.

## 4 Crossing direction analysis

In the last section, we have described a process of determining the set of  $\tau$  for the system to be asymptotically stable within the given interval of interest  $\mathcal{T} = [\tau^l, \tau^u]$ 

provided that  $\text{Inc}(\omega^*, \tau^*)$  can be determined for any given critical pair  $(\omega^*, \tau^*)$ . A critical pair is one that satisfies

$$D(j\omega^*, \tau^*) = 0. \tag{51}$$

In general,  $\text{Inc}(\omega^*, \tau^*)$  may be determined by a numerical method. It is, however, obviously of interest to use an analytical method. This is the focus of this section. For this purpose, we first replace Assumption III by one that can be easily verified analytically.

Assumption IIIa. The pair  $(\omega^*, \tau^*)$  satisfies (51),  $D(\lambda, \tau)$  is differentiable with respect to  $\tau$  up to Nth order in a neighborhood of  $(j\omega^*, \tau^*)$ . Furthermore,

$$\partial_{\omega} F(\omega^*, \tau^*) \neq 0. \tag{52}$$

It should be pointed out that Assumption IIIa is not equivalent to Assumption III. For the single delay case, it was shown in Lemma 1 of [20] that the last part of Assumption IIIa implies the first part of Assumption III. The following lemma shows that this is also true for the commensurate delay case.

**Lemma 3.** Suppose  $(\omega^*, \tau^*)$  satisfies (51), and  $D(\lambda, \tau)$  is differentiable with respect to  $\tau$  in a neighborhood of  $(j\omega^*, \tau^*)$ . If it satisfies the condition (52) in Assumption IIIa, then it also satisfies the condition (31) in Assumption III.

*Proof.* It is sufficient to show that if (31) is not satisfied, then (52) is not satisfied. In other words, we only need to show that

$$\partial_{\lambda} D(j\omega^*, \tau^*) = 0, \tag{53}$$

implies

$$\partial_{\omega} F(\omega^*, \tau^*) = 0. \tag{54}$$

Let (53) be satisfied. First, assume  $P_M(j\omega^*, \tau^*) \neq 0$ . Consider the decomposition equation

$$\Phi_{\omega\tau^*}(x) = \sum_{k=0}^{M} P_k(j\omega, \tau^*) x^k = 0.$$
 (55)

The equation (51) means that, for  $\omega = \omega^*$ , the polynomial equation (55) has a solution  $x^* = e^{-j\omega^*\tau^*}$ . If  $x^*$  is a multiple root of  $\Phi_{\omega^*\tau^*}(x)$ , then (54) is satisfied according to Theorem 1. It remains to be shown that (54) is satisfied when  $x^*$  is a simple root of  $\Phi_{\omega^*\tau^*}(x)$ .

For  $\omega$  sufficiently close to  $\omega^*$ , let

$$\varepsilon = -D(j\omega, \tau^*) = -\Phi_{\omega\tau^*}(e^{-j\omega\tau^*}).$$
(56)

Then, (51) and (53) imply that  $j\omega^*$  is a multiple root of  $D_{\tau^*}(j\omega)$ . It follows that

$$\lim_{\omega \to \omega^*} \frac{\varepsilon}{\omega - \omega^*} = \frac{d\varepsilon}{d\omega} \bigg|_{\omega = \omega^*} = -j \frac{dD_{\tau^*}}{d\lambda} \bigg|_{\lambda = j\omega^*} = 0.$$
(57)

Equation (56) can be written as

$$P_0(j\omega,\tau^*) + \varepsilon + \sum_{k=1}^M P_k(j\omega,\tau^*)\hat{x}^k = 0, \qquad (58)$$

where

$$\hat{x} = e^{-j\omega\tau^*}.$$
(59)

Comparing the equations (58) and (55), the two polynomial equations differ by a small parameter  $\varepsilon$  in the constant term. Therefore, there is a solution of (55)

$$x = \hat{x} + \delta,\tag{60}$$

where  $\delta$  is small. Furthermore, as x is a simple root of  $\Phi_{\omega\tau^*}(x)$ ,  $\delta$  is also differentiable with respect to  $\varepsilon$ . Therefore

$$\lim_{\varepsilon \to 0} \left| \frac{\delta}{\varepsilon} \right| < \infty.$$
(61)

Applying Lemma 2 to  $\Phi_{\omega^*\tau^*}(x^*)$  and  $\Phi_{\omega\tau^*}(x)$ , we obtain

$$F(\omega^*, \tau^*) = 0, \tag{62}$$

and

$$F(\omega, \tau^*) = Q(\omega, \tau^*)(1 - x\bar{x}), \tag{63}$$

where  $Q(\omega, \tau^*)$  is continuous. But

$$1 - x\bar{x} = 1 - (\hat{x} + \delta)(\hat{x} + \delta)$$
  
=  $1 - |\hat{x}|^2 - \hat{x}\overline{\delta} - \overline{(\hat{x}\overline{\delta})} - \delta\overline{\delta}$   
=  $-\hat{x}\overline{\delta} - \overline{(\hat{x}\overline{\delta})} - \delta\overline{\delta}.$  (64)

Therefore,

$$\partial_{\omega}F(\omega^*,\tau^*) = \lim_{\omega \to \omega^*} \frac{-Q(\omega,\tau)\left[\hat{x}\bar{\delta} + \overline{(\hat{x}\bar{\delta})} + \delta\bar{\delta}\right]}{\varepsilon} \frac{\varepsilon}{\omega - \omega^*}$$
  
= 0 (65)

in view of (61) and (57). In other words, (54) is satisfied.

If  $P_M(j\omega^*, \tau^*) = 0$ , then  $P_0(j\omega^*, \tau^*) \neq 0$  due to Assumption II. We may consider  $\Psi_{\omega\tau^*}(x)$  instead of  $\Phi_{\omega\tau^*}(x)$ , and the proof proceeds similarly.

Under Assumption IIIa, the above lemma indicates that the equation

$$D(\lambda,\tau) = 0 \tag{66}$$

defines a unique function  $\lambda(\tau)$  when  $\tau$  is in a sufficiently small neighborhood of  $\tau^*$ , and  $\lambda$  is restricted to a small neighborhood of  $j\omega^*$ . This function is differentiable up to Nth order. Furthermore, if for some  $m \leq N$ ,

$$\mathcal{R}\left(\frac{d^k\lambda(\tau^*)}{d\tau^k}\right) = 0, \quad k = 1, 2, \dots m - 1,$$
(67)

$$\mathcal{R}\left(\frac{d^m\lambda(\tau^*)}{d\tau^m}\right) \neq 0,\tag{68}$$

and  $\tau^* > \tau^l$ , then

$$\operatorname{Inc}(\omega^*, \tau^*) = \begin{cases} \operatorname{sgn}\left(\mathcal{R}\left(\frac{d^m \lambda(\tau^*)}{d\tau^m}\right)\right), \text{ if } m \text{ is odd,} \\ 0, & \text{ if } m \text{ is even.} \end{cases}$$
(69)

If  $\tau^* = \tau^l$  instead, then

$$\operatorname{Inc}(\omega^*, \tau^*) = \max\left\{0, \operatorname{sgn}\left(\mathcal{R}\left(\frac{d^m\lambda(\tau^*)}{d\tau^m}\right)\right)\right\}.$$
(70)

In view of the above discussions, calculating m and  $\operatorname{sgn}\left(\mathcal{R}\left(\frac{d^m\lambda(\tau^*)}{d\tau^m}\right)\right)$  is critical in the determination of the crossing direction of characteristic roots. In [20], the case with single delay and m = 1 was solved. The following theorem generalizes the result to the commensurate delay case with  $m \geq 1$ .

**Theorem 2.** Let Assumption IIIa be satisfied. For all  $\tau$  in a sufficiently small neighborhood of  $\tau^*$ , let  $(\omega(\tau), \theta(\tau))$  be the unique solution in a small neighborhood of  $(\omega^*, 2r\pi)$  to the equation

$$\Phi(\omega,\tau,e^{j(\theta-\omega\tau)}) = 0, \tag{71}$$

and let  $\lambda(\tau)$  be the unique solution in a small neighborhood of  $j\omega^*$  to the equation (66). If

$$\frac{d^k \theta(\tau^*)}{d\tau^k} = 0, \quad k = 1, 2, \dots m - 1,$$
(72)

$$\frac{d^m \theta(\tau^*)}{d\tau^m} \neq 0,\tag{73}$$

then,

$$\left(\mathcal{R}\left(\frac{d^k\lambda(\tau^*)}{d\tau^k}\right)\right) = 0, \quad k = 1, 2, \dots m - 1,$$
(74)

$$\operatorname{sgn}\left(\mathcal{R}\left(\frac{d^m\lambda(\tau^*)}{d\tau^m}\right)\right) = (-1)^{N_x(\tau^*)}\operatorname{sgn}\left(\frac{\partial F(\omega^*,\tau^*)}{\partial\omega}\right)\operatorname{sgn}\left(\frac{d^m\theta(\tau^*)}{d\tau^m}\right),\tag{75}$$

where  $N_x(\tau)$  is the number of roots of the polynomial  $\Phi_{\omega(\tau)\tau}(x)$  outside the unit circle if  $P_M(j\omega(\tau),\tau) \neq 0$ , or the number of roots of the polynomial  $\Psi_{\omega(\tau)\tau}(x)$  inside the unit circle if  $P_0(j\omega(\tau),\tau) \neq 0$ .

The proof of this theorem will be given in Appendix D. The theorem indicates that if the devatives of  $\theta(\tau)$  vanish up to the (m-1)th order, then the real part of the derivatives of the critical characteristic roots also vanish up to the (m-1)th order. Furthermore, the sign of the lowest-order nonzero derivative of  $\theta(\tau)$  can be related to sign of the real part of the lowest-order derivative of critical characteristic root with nonzero real part. This means that the crossing direction of a critical characteristic root can be obtained by the crossing direction of the phase curve  $\theta(\tau)$ , which is formally stated in the following corollary.

**Corollary 2.** Let  $\tau^* \in \mathcal{T}_o^{(i)} = (\tau^{(i-1)}, \tau^{(i)})$ , and  $m \leq N$ . If  $\theta(\tau)$  moves from below the  $2r\pi$  line to above it as  $\tau$  increases through  $\tau^*$ , then,  $\operatorname{Inc}(\omega^*, \tau^*) = (-1)^{N_x(\tau^*)} \operatorname{sgn}\left(\frac{\partial F(\omega^*, \tau^*)}{\partial \omega}\right)$ . If  $\theta(\tau)$  crosses the  $2r\pi$  line from above to below instead, then  $\operatorname{Inc}(\omega^*, \tau^*) = (-1)^{N_x(\tau^*)+1} \operatorname{sgn}\left(\frac{\partial F(\omega^*, \tau^*)}{\partial \omega}\right)$ .

*Proof.* This is obvious from (75) in Theorem 2 and (69).

The above corollary means that if the curve  $\theta(\tau)$  generated in Section III is sufficiently smooth, then the crossing direction (and therefore  $\operatorname{Inc}(\omega^*, \tau^*)$ ) can be obtained by observing the direction that  $\theta(\tau)$  crosses the  $2r\pi$  line, provided that  $N_x(\tau^*)$  and the sign of  $\frac{\partial F(\omega^*, \tau^*)}{\partial \omega}$  are available. The following proposition means that these two values need to be checked only at one point in each continuous frequency curve in each CFCI. **Proposition 1.** Within each  $\mathcal{T}_o^{(i)}$  and for each continuous frequency curve  $\omega(\tau)$ that satisfies  $F(\omega(\tau), \tau) = 0$  defined in  $\mathcal{T}_o^{(i)}$ ,  $\operatorname{sgn}\left(\frac{\partial F(\omega, \tau)}{\partial \omega}\Big|_{\omega=\omega(\tau)}\right)$  and  $N_x(\tau)$  remain constant.

*Proof.* By the definition of  $\mathcal{T}_o^{(i)}$ ,

$$\frac{\partial F(\omega(\tau),\tau)}{\partial \omega} \neq 0, \quad \tau \in \mathcal{T}_o^{(i)}.$$
(76)

To show sign invariance of  $\frac{\partial F(\omega,\tau)}{\partial \omega}\Big|_{\omega=\omega(\tau)}$ , observe that  $\frac{\partial F(\omega,\tau)}{\partial \omega}\Big|_{\omega=\omega(\tau)}$  is a continuous function of  $\tau$ , and its sign can change only if it vanishes first, which contradicts (76).

To show the invariance of  $N_x(\tau)$ , observe that  $\Phi_{\omega^*\tau^*}(x)$  has  $N_x(\tau^*)$  roots outside of the unit circle, one root on the unit circle, and  $M - N_x(\tau^*) - 1$  roots inside the unit circle if  $P_M(j\omega(\tau), \tau) \neq 0$ . In order for  $N_x(\tau)$  to change within  $\mathcal{T}_o^{(i)}$ , due to continuity of the roots of  $\Phi_{\omega(\tau)\tau}(x)$  with respect to  $\tau$  when  $P_M(j\omega(\tau), \tau) \neq 0$ , either one of roots outside the unit circle, or one of the roots inside the unit circle must first reach the unit circle for some  $\tau = \hat{\tau} \in \mathcal{T}_o^{(i)}$ . But this would mean that  $\Phi_{\omega(\hat{\tau})\hat{\tau}}(x)$  has two roots at the unit circle, which would require  $\frac{\partial F(\omega,\hat{\tau})}{\partial \omega}\Big|_{\omega=\omega(\hat{\tau})} = 0$  in view of Theorem 1, thus contradicting (76).

At a  $\tau$  such that  $P_M(j\omega(\tau), \tau) = 0$ , then  $P_0(j\omega(\tau), \tau) \neq 0$  per Assumption II, the invariance of  $N_x(\tau)$  can be shown by considering  $\Psi_{\omega(\tau)\tau}(x)$  instead of  $\Phi_{\omega(\tau)\tau}(x)$ .  $\Box$ 

It is also possible to analytically obtain  $\frac{d^k \theta(\tau)}{d\tau^k}$ , k = 1, 2, ...m. This can be done by taking the derivatives of the equation  $\Phi(\omega, \tau, e^{j(\theta - \omega \tau)}) = 0$ ,

$$\frac{d^k}{d\tau^k} \Phi(\omega, \tau, e^{j(\theta - \omega\tau)})|_{\tau = \tau^*} = 0,$$
(77)

and solving for  $\frac{d^k \theta(\tau^*)}{d\tau^k}$  and  $\frac{d^k \omega(\tau^*)}{d\tau^k}$ . The nonzero  $\frac{d^m \theta(\tau^*)}{d\tau^m}$  may be obtained by a recursive process, although the formulation is progressively more tedious as *m* increases. In Appendix C, it will be shown that this process is always possible and more specifics of this solution process are provided there.

## 5 Examples

In this section, we will present two examples to illustrate the stability analysis using the results developed in this paper. The coefficients polynomials depend on the delay either in poynomial or exponential form so that it is easier to see some of the steps analytically. In practice, these may be any other functions as long as the continuity or differentiability conditions are satisfied. We also present an example of stability analysis when some of the nondegeneracy assumptions are violated.

**Example 4.** Consider the characteristic equation (9). By definition,

$$P_0 = \lambda - \alpha, \ P_1 = e^{\alpha \tau}, \ P_2 = e^{2\alpha \tau}.$$

Set the delay interval of interest  $\mathcal{T} = [0, 0.8]$  and  $\alpha = \frac{3}{2}$ . The expression of function F can be obtained using (18) as

$$F = -\omega^4 + a_1\omega^2 + a_2,$$

where

$$a_{1} = 2a^{4} + a^{2} - \frac{9}{2},$$
  

$$a_{2} = -a^{8} + a^{6} + \frac{15}{2}a^{4} + \frac{9}{4}a^{2} - \frac{81}{16}$$
  

$$a = e^{\frac{3}{2}\tau}.$$

By solving (23) and (33) for  $(\omega, \tau) \in \mathbb{R}_+ \times \mathcal{T}$ , we find that  $\mathcal{T}$  can be decomposed into two CFCIs  $\mathcal{T}^{(1)} = [\tau^{(0)}, \tau^{(1)}]$  and  $\mathcal{T}^{(2)} = [\tau^{(1)}, \tau^{(2)}]$ , where  $\tau^{(0)} = 0$ ,  $\tau^{(1)} \approx 0.4045$ ,  $\tau^{(2)} = 0.8$ . Polynomial  $F_{\tau}(\omega)$  has one real solution for  $\tau \in \mathcal{T}_0^{(1)}$  and two real solutions for  $\tau \in \mathcal{T}_0^{(2)}$ . Consequently the frequency curves  $\omega_1^{(1)}(\tau)$ ,  $\omega_1^{(2)}(\tau)$ ,  $\omega_2^{(2)}(\tau)$  are well defined in the corresponding CFCIs as plotted in Fig. 1a. The associated phase curves are plotted in Fig. 1b. The graph of  $\theta_1^{(1)}$  crosses the horizontal line 0 at  $\tau_1 \approx 0.2368$  and the graphs of  $\theta_2^{(2)}$  and  $\theta_1^{(2)}$  cross the horizontal line  $2\pi$  at  $\tau_2 \approx 0.6878$  and  $\tau_3 \approx 0.6976$ respectively, all from below to above. By definition, we have  $\omega_{11} = \omega_1^{(1)}(\tau_1) \approx 2.9010$ ,  $\omega_{21} = \omega_2^{(2)}(\tau_2) \approx 5.4195$ ,  $\omega_{31} = \omega_1^{(2)}(\tau_3) \approx 10.0461$ .

We verify that the following holds:

$$\begin{aligned} \partial_{\omega} F(\omega_{1}^{(1)}(\tau),\tau) < 0, \ N_{x}(\omega_{1}^{(1)}(\tau),\tau) &= 1, \ \forall \tau \in \mathcal{T}^{(1)}, \\ \partial_{\omega} F(\omega_{1}^{(2)}(\tau),\tau) > 0, \ N_{x}(\omega_{1}^{(2)}(\tau),\tau) &= 0, \ \forall \tau \in \mathcal{T}^{(2)}, \\ \partial_{\omega} F(\omega_{2}^{(2)}(\tau),\tau) > 0, \ N_{x}(\omega_{2}^{(2)}(\tau),\tau) &= 0, \ \forall \tau \in \mathcal{T}^{(2)}. \end{aligned}$$

Therefore, we conclude from (75) that  $Inc(\omega^*, \tau^*) = 1$  at all the three crossing points. In other words, the imaginary roots  $j\omega_{11}$ ,  $j\omega_{21}$ ,  $j\omega_{31}$  all move toward the RHP as the delay value increases and sweeps through  $\tau_1$ ,  $\tau_2$ ,  $\tau_3$ , respectively. It can be verified that  $N^u(0) = 0$ , thus the number of unstable roots can be easily computed using (49) and is plotted against  $\tau$  in Fig. 1c. It can thus be concluded that the system (8) is  $\alpha$ -stable with  $\alpha = 1.5$  for  $\tau \in [0, \tau_1)$  and not  $\alpha$ -stable for  $\tau \in [\tau_1, 0.8]$ .

**Example 5.** Consider the characteristic equation (1) with the delay interval of interest  $\mathcal{T} = [0, 0.65]$  and the following  $\tau$ -dependent coefficient polynomials:

$$P_0 = \lambda^2 + 4, \ P_1 = (1 - 4e^{-2\tau}), \ P_2 = (1 - 2e^{-2\tau})\lambda.$$

The frequency curves  $\omega_k^{(i)}(\tau)$  are plotted in Fig. 2a. The associated phase curves  $\theta_k^{(i)}$  are plotted in Fig. 2b. The curves  $\theta_2^{(1)}$  and  $\theta_3^{(2)}$  cross the horizontal line 0 at  $\tau_1 \approx 0.1906$  and  $\tau_2 \approx 0.5243$ , respectively, both from below to above. The following can be checked:

$$\begin{aligned} \partial_{\omega} F(\omega_{2}^{(1)}(\tau),\tau) &> 0, \ N_{x}(\omega_{2}^{(1)}(\tau),\tau) = 1, \ \forall \tau \in \mathcal{T}^{(1)}.\\ \partial_{\omega} F(\omega_{3}^{(2)}(\tau),\tau) &> 0, \ N_{x}(\omega_{3}^{(2)}(\tau),\tau) = 0, \ \forall \tau \in \mathcal{T}^{(2)}. \end{aligned}$$

Therefore, we can conclude from equation (75) that  $Inc(\omega_2^{(1)}(\tau_1), \tau_1) = -1$  and  $Inc(\omega_3^{(2)}(\tau_2), \tau_2) = 1$ . It can be verified that  $N^u(0) = 2$ , then the number of unstable roots can be easily computed using (49) and is plotted against  $\tau$  in Fig. 2c. We can therefore conclude that this system is stable for  $\tau \in (\tau_1, \tau_2)$ .

**Example 6.** This example illustrates a degenerate case. Consider the characteristic equation (1) with

$$P_0 = \lambda^2, P_1 = \lambda + \tau, P_2 = \lambda, P_3 = \lambda,$$

and the delay interval of interest is  $\mathcal{T} = [0,3]$ . The function  $F(\omega,\tau)$  can be calculated as

$$F = \omega^{12} - 3\omega^{10} + (2\tau - 4\tau^2)\omega^8 + 9\tau^2\omega^6 - 12\tau^3\omega^4 + 4\tau^4\omega^2.$$

Obviously, the equations (23) and (33) are satisfied for all  $\tau \in \mathcal{T}$  and  $\omega = 0$ , which violates Assumption IV. However, the principles presented in this paper can still be used to analyze this system.

For  $\tau = \tau^l = 0$ ,  $D_{\tau}(\lambda)$  has two roots, -1 and 0. The root at 0 moves to the LHP as  $\tau$  increases from 0 because

$$\frac{d\lambda}{d\tau}\Big|_{\lambda=0,\tau=0} = -\frac{\frac{\partial D}{\partial\tau}}{\frac{\partial D}{\partial\lambda}}\Big|_{\lambda=0,\tau=0} = -1.$$



**Fig. 1** Plots for Example 4: (a) Frequency curves: There is one frequency curve,  $\omega_1^{(1)}(\tau)$ , in  $\mathcal{T}^1$ ; and there are two frequency curves,  $\omega_1^2(\tau)$  and  $\omega_2^2(\tau)$ , in  $\mathcal{T}^2$ . (b) The corresponding phase curves:  $\theta_k^{(i)}(\tau)$  corresponds to the frequency curve  $\omega_k^{(i)}(\tau)$ . (c) The number of roots on the right half-plane,  $N^u(\tau)$ , for  $\tau \in \mathcal{T}$ .



**Fig. 2** Plots for Example 5: (a) Frequency curves: There are two frequency curves,  $\omega_k^{(1)}(\tau), k = 1, 2$ in  $\mathcal{T}^1$ ; and there are four frequency curves,  $\omega_k^{(2)}(\tau), k = 1, 2, 3, 4$ , in  $\mathcal{T}^2$ . (b) The corresponding phase curves:  $\theta_k^{(i)}(\tau)$  corresponds to the frequency curve  $\omega_k^{(i)}(\tau)$ . (c) The number of roots on the right halfplane,  $N^u(\tau)$ , for  $\tau \in \mathcal{T}$ .

Therefore,  $\operatorname{Inc}(\tau_1) = 0$  per the equation (70), and the system is stable for  $\tau$  sufficiently small. We may still identify all the crossing points by following the same process as described in Section III. Indeed, for  $\tau$  in the entire interval  $\mathcal{T}$ , there are three real solutions of  $\omega_i(\tau)$ , i = 1, 2, 3, and they are plotted in Fig. 3a. For  $\omega_3(\tau) = 0$ , it can be easily calculated that  $\Phi_{\omega\tau}(x)$  has two roots on the unit circle,  $x = \pm j$ . The corresponding  $\theta_{3\pm} = \pm \pi/2$ . As they do not touch any  $r2\pi$  line, we can conclude that  $\lambda = 0$  is not a root for  $\tau > 0$ , which can also be directly verified. For  $\omega_1(\tau)$  and  $\omega_2(\tau)$ ,  $\Phi_{\omega\tau}(x)$  has a single root on the unit circle, and the corresponding  $\theta_1(\tau)$  and  $\theta_2(\tau)$ are plotted in Fig. 3b along with  $\theta_{3\pm}(\tau)$ . It can be seen that  $\theta_1(\tau)$  crosses 0 line at  $\tau_2 \approx 0.5382$ , and then the  $2\pi$  line at  $\tau_3 \approx 2.7206$ , both from below to above. It can be shown that Theorem 2 still applies, and it can be calculated that  $\partial_{\omega}F(\omega_1(\tau), \tau) > 0$  and  $N_x(\omega_1(\tau), \tau) = 2$  at both  $\tau = \tau_2$  and  $\tau_3$ . Therefore,  $\operatorname{Inc}(\omega_1(\tau_2), \tau_2) = \operatorname{Inc}(\omega_1(\tau_3), \tau_3) =$ 1, and  $N^u(\tau)$  is plotted in Fig. 3c, from which we conclude that the system is stable for  $\tau = (\tau_1, \tau_2) = (0, 0.5382)$ .

## 6 Conclusions and Future Research

A method to analyze the stability of linear time-delay systems represented by differential-difference equations with commensurate delays and delay-dependent coefficients is developed. This method generalizes the previous results for the single delay case. The crossing direction analysis has also been extended to the case that requires higher-order derivatives. The method takes advantage of Orlando's formula. Under some nondegeneracy assumptions parallel to the single delay case, it was shown that the decomposition polynomial has one and only one solution on the unit circle, which allows us to develop a method of systematic stability analysis parallel to the single delay case.

To explore further along this line of research, it is interesting to carry out stability analysis described in this paper for practical systems. Some practical issues such as complexity and amount of calculation may arise. While the assumptions given in this paper are "generic", it would be interesting to see whether these assumptions may be violated in practice, and extensions such as covered in Example 6 may be desirable.

It is also interesting to extend the results discussed here to multiple group of commensurate delays. Such a problem is rather challenging to solve. However, even some special cases would be very interesting. Indeed, when there are two groups with each group consisting of a single delay, a special case was solved in [1].

## Appendices

In the appendices, we state the Orlando's formula in Appendix A, which forms an important basis of this paper. Appendix B gives a brief discussion on the derivatives of multivariable functions with dependence between variables. Then we will present a procedure of calculating the derivatives of phase angle with respect to  $\tau$ , along with some important observations in Appendix C. Finally, we will prove Theorem 2 in Appendix D.



**Fig. 3** Plots for Example 6, which is a degenerate case: (a) Frequency curves: There three frequency curves,  $\omega_k(\tau), k = 1, 2, 3$  in  $\mathcal{T}$ , which is not divided in this case. (b) The corresponding phase curves:  $\theta_1(\tau)$  and  $\theta_2(\tau)$  correspond to  $\omega_1(\tau)$  and  $\omega_2(\tau)$ . Two phase curves,  $\omega_{3+}(\tau)$  and  $\omega_{3-}(\tau)$  correspond to the frequency curve  $\omega_3(\tau)$ . (c) The number of roots on the right half-plane,  $N^u(\tau)$ , for  $\tau \in \mathcal{T}$ .

# Appendix A Orlando's formula

For a polynomial

$$\Gamma(x) = \sum_{k=0}^{M} a_k x^k, \tag{A1}$$

 $a_M \neq 0$ , let

$$\bar{\Gamma}(x) = \sum_{k=0}^{M} \bar{a}_{M-k} x^k, \tag{A2}$$

and

$$\Xi = \bar{\Gamma}^{H}(S)\bar{\Gamma}(S) - \Gamma^{H}(S)\Gamma(S), \qquad (A3)$$

where the superscript H represents Hermitian transpose, and S is an  $M\times M$  shift matrix

$$S = \begin{pmatrix} 0 & 1 & 0 & \cdots & 0 \\ 0 & 0 & 1 & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \cdots & 1 \\ 0 & 0 & 0 & \cdots & 0 \end{pmatrix}.$$
 (A4)

Then the following equation is known as the Orlando's formula [22, 33]

$$\det(\Xi) = |a_M|^{2M} \prod_{i=1}^{M} \prod_{j=1}^{M} (1 - x_i \bar{x}_j),$$
(A5)

where  $x_i$ , i = 1, 2, ..., M are the roots of the polynomial (A1). An immediate implication of the Orlando's formula is that the polynomial (A1) has a root on the unit circle only if det( $\Xi$ ) = 0. This implication is critical in the stability analysis of systems with commensurate delays (with coefficients independent of delays)[6, 7, 18, 19]. Indeed, in stability analysis of such systems, the coefficients  $a_k$  become polynomials  $P_k(j\omega)$  of frequency. Setting det( $\Xi$ ) to zero allows one to identify the crossing frequencies, one may identify the corresponding critical delays from the roots of  $\Gamma(x)$  on the unit circle.

## Appendix B Derivatives of multivariable functions

The following lemma is useful in the proof of a number of results in this paper. Lemma 4. Let

$$h(x) = f(x)g(x)$$

where f(x) is continuous and g(x) is continously differentiable in a neighborhood of  $x^*$ , and

$$g(x^*) = 0.$$
 (B6)

Then, h(x) is differentiable at  $x^*$ , and

$$h'(x^*) = f(x^*)g'(x^*).$$
 (B7)

*Proof.* By definition of derivative and using (B6), we obtain

$$h'(x^*) = \lim_{x \to x^*} \frac{f(x)g(x)}{x - x^*} = \lim_{x \to x^*} f(x)\frac{g(x) - g(x^*)}{x - x^*}$$

from which (B7) is obvious.

We start with making some observations of derivatives of multivariable functions with variable dependence. The functions are assumed to be continuously differentiable to a sufficiently high order.

**Lemma 5.** Consider a function  $g(x, \tau)$  of two variables x and  $\tau$ . If x depends on  $\tau$ , then

$$\frac{d^{\kappa}g}{d\tau^{k}} = \frac{\partial g}{\partial x}x^{(k)}(\tau) + \frac{\partial^{\kappa}g}{\partial\tau^{k}} + U_{k},$$
(B8)

where  $U_k = U_k(x,\tau;x',x'',\ldots,x^{(k-1)})$  is a polynomial of  $x'(\tau), x''(\tau),\ldots, x^{(k-1)}(\tau)$ that satisfies

$$U_k(x,\tau;0,0,\ldots,0) = 0.$$
 (B9)

The coefficients of the polynomial  $U_k$ , which are functions of x and  $\tau$ , are linear combinations of partial derivatives

$$\frac{\partial^{p+q}g}{\partial x^p \partial \tau^q}, \quad 1 \le p; 0 \le q; 2 \le p+q \le k.$$
(B10)

The above lemma can be easily shown by induction. The explicit expression of  $U_k$  may also be obtained for a small k, for example,

$$U_2 = \frac{\partial^2 g}{\partial x^2} \left[ x'(\tau) \right]^2 + 2 \frac{\partial^2 g}{\partial x \partial \tau} x'(\tau), \tag{B11}$$

although the expression becomes progressively more tedious as k increases.

Consider now a function of three variables  $h(x, y, \tau)$ . When y depends on  $\tau$ , it becomes a function of two variables x and  $\tau$ . The (p+q)th order partial derivative of the function with respect to x and  $\tau$  considering this dependence is denoted as  $\frac{\partial_{y(t)}^{p+q}h}{\partial x^{p}\partial \tau^{q}}$ . For example,

$$\begin{split} \frac{\partial_{y(\tau)}h}{\partial\tau} &= \frac{\partial h}{\partial y}y'(\tau) + \frac{\partial h}{\partial\tau},\\ \frac{\partial_{y(\tau)}^2h}{\partial x\partial\tau} &= \frac{\partial^2 h}{\partial x\partial y}y'(\tau) + \frac{\partial^2 h}{\partial x\partial\tau},\\ \frac{\partial_{y(\tau)}^2h}{\partial\tau^2} &= \frac{\partial^2 h}{\partial y^2}\left[y'(\tau)\right]^2 + 2\frac{\partial^2 h}{\partial y\partial\tau}y'(\tau) + \frac{\partial h}{\partial y}y''(\tau) + \frac{\partial^2 h}{\partial\tau^2} \end{split}$$

With the above notation, Lemma 5 may be applied to a function of three variables to obtain the following corollary.

27

**Corollary 3.** Consider a function  $h(x, y, \tau)$ . If x and y depend on  $\tau$ , then

$$\frac{d^k h}{d\tau^k} = \frac{\partial h}{\partial x} x^{(k)}(\tau) + \frac{\partial_{y(\tau)}^k h}{\partial \tau^k} + U_k, \tag{B12}$$

where  $U_k = U_k(x, y, \tau; x'(\tau), x''(\tau), \dots, x^{(k-1)}(\tau))$  is a polynomial of  $x', x'', \dots, x^{(k-1)}$ that satisfies

$$U_k(x, y, \tau; 0, 0, \dots, 0) = 0.$$
 (B13)

Furthermore, the term  $\frac{\partial_{y(\tau)}^{k}h}{\partial\tau^{k}}$  may be expressed as

$$\frac{\partial_{y(\tau)}^{k}h}{\partial\tau^{k}} = \frac{\partial h}{\partial y}y^{(k)}(\tau) + \frac{\partial^{k}h}{\partial\tau^{k}} + V_{k}, \tag{B14}$$

where  $V_k = V_k(x, y, \tau; y', y'', \dots, y^{(k-1)})$  is a polynomial of  $y', y'', \dots, y^{(k-1)}$  that satisfies

$$V_k(x, y, \tau; 0, 0, \dots, 0) = 0.$$
 (B15)

# Appendix C Derivatives of phase angle

With the notation in Appendix B, we will now discuss the calculation of the derivative  $\frac{d^k\theta}{d\tau^k}$  in a neighborhood of  $\tau^*$ . For this purpose, define

$$T(\theta, \omega, \tau) = \Phi(\omega, \tau, e^{j(\theta - \omega\tau)}), \tag{C16}$$

$$W(\omega,\tau) = \Phi(\omega,\tau,e^{-j\omega\tau}).$$
(C17)

Then  $\theta(\tau)$  and  $\omega(\tau)$  satisfy  $\theta(\tau^*) = 2r\pi$ ,  $\omega(\tau^*) = \omega^*$ , and

$$T(\theta, \omega, \tau) = 0. \tag{C18}$$

Taking kth order derivative of the above equation and applying Corollary 3, yield

$$\left(\frac{d^{k}T}{d\tau^{k}}\right)_{*} = \left(\frac{\partial T}{\partial \theta}\right)_{*} \left(\frac{d^{k}\theta}{d\tau^{k}}\right)_{*} + \left(\frac{\partial W}{\partial \omega}\right)_{*} \left(\frac{d^{k}\omega}{d\tau^{k}}\right)_{*} + \left(V_{Wk}\right)_{*} + \left(U_{Tk}\right)_{*} = 0,$$
(C19)

where  $(\cdot)_*$  means that the expression is evaluated at  $\tau = \tau^*, \omega = \omega^*$  and  $\theta = 2r\pi$ ,  $V_{Wk}$  is a polynomial of  $\omega'(\tau), \omega''(\tau), \ldots, \omega^{(k-1)}(\tau)$ , and  $U_{Tk}$  is a polynomial of  $\theta'(\tau), \theta''(\tau), \ldots, \theta^{(k-1)}(\tau)$ . In arriving at the above expression, we have used the fact that  $T(\omega, \tau)|_{\theta=2r\pi} = W(\omega, \tau)$ .

Suppose

$$\left(\frac{d^i\theta}{d\tau^i}\right)_* = 0, \ i = 1, 2, \dots, k-1.$$
(C20)

Then

$$(U_{Tk})_* = 0.$$
 (C21)

Therefore, we can solve (C19) for  $\left(\frac{d^k\theta}{d\tau^k}\right)_*$  by multiplying  $\overline{\left(\frac{\partial W}{\partial\omega}\right)}_*$  and taking the imaginary part

$$\left(\frac{d^{k}\theta}{d\tau^{k}}\right)_{*} = -\frac{\mathcal{I}\left(\left(\frac{\partial W}{\partial\omega}\right)_{*}\left(V_{Wk}\right)_{*}\right)}{\mathcal{I}\left(\overline{\left(\frac{\partial W}{\partial\omega}\right)}_{*}\left(\frac{\partial T}{\partial\theta}\right)_{*}\right)}.$$
(C22)

Similarly,  $\left(\frac{d^k\omega}{d\tau^k}\right)_*$  may be solved from (C19) by multiplying  $\overline{\left(\frac{\partial T}{\partial \theta}\right)}_*$  and taking the imaginary part

$$\left(\frac{d^{k}\omega}{d\tau^{k}}\right)_{*} = -\frac{\mathcal{I}\left(\left(\frac{\partial T}{\partial \theta}\right)_{*}\left(V_{Wk}\right)_{*}\right)}{\mathcal{I}\left(\left(\frac{\partial W}{\partial \omega}\right)_{*}\overline{\left(\frac{\partial T}{\partial \theta}\right)_{*}}\right)}.$$
(C23)

The equations (C22) and (C23) may be used starting from k = 1 and increasing k until we reach

$$\left(\frac{d^{\kappa}\theta}{d\tau^k}\right)_* \neq 0,$$

which means that m in Theorem 2 has been reached.

In order for the equations (C22) and (C23) to be valid, the denominator needs to be nonzero. In order to prove this, we need the following lemma.

**Lemma 6.** Let Assumption IIIa be satisfied, and let  $x_1(\omega)$  be a root of  $\Phi_{\omega\tau^*}(x)$  in a small neighborhood of  $x_1^* = e^{-j\omega^*\tau^*}$  for  $\omega$  in a small neighborhood of  $\omega^*$ . Then  $x_1(\omega)$  is differentiable and satisfies

$$\mathcal{R}(x_1'(\omega^*)\bar{x}_1^*) \neq 0. \tag{C24}$$

Furthermore,

$$\operatorname{sgn}\left(\mathcal{R}(x_1'(\omega^*)\bar{x}_1^*)\right) = (-1)^{N_x(\tau^*)}\operatorname{sgn}\left(\partial_\omega F(\omega^*,\tau^*)\right).$$
(C25)

*Proof.* Assume  $P_M(j\omega^*, \tau^*) \neq 0$ . Then, we can write

$$F(\omega, \tau^*) = R(\omega)(1 - x_1(\omega)\bar{x}_1(\omega)), \qquad (C26)$$

where

$$R(\omega) = -|P_M(j\omega,\tau)|^{2M} \left(\prod_{i=2}^M \chi_{ii}\right) \left(\prod_{k=1}^{M-1} \prod_{i=k+1}^M \chi_{ki}\bar{\chi}_{ki}\right), \quad (C27)$$
  
$$\chi_{ki} = 1 - x_k(\omega)\bar{x}_i(\omega), \quad k, i = 1, 2, \dots, M,$$

and  $x_k(\omega)$ , k = 2, 3, ..., M are the remaining M - 1 roots of  $\Phi_{\omega\tau^*}(x)$ . Applying Theorem 1, we conclude that  $x_1(\omega)$  is differentiable,  $R(\omega)$  is continuous and satisfies

$$R(\omega^*) \neq 0. \tag{C28}$$

Furthermore, it is easy to see from (C27) that

$$\operatorname{sgn}(R(\omega^*)) = (-1)^{N_x(\tau^*)+1}.$$
(C29)

Taking derivative of (C26) and evaluating it at  $\omega = \omega^*$  using Lemma 4 yield

$$\partial_{\omega} F(\omega^*, \tau^*) = -R(\omega^*)(x_1'(\omega^*)\bar{x}_1^* + x_1^* \overline{x_1'(\omega^*)})$$
$$= -R(\omega^*)2\mathcal{R}(x_1'(\omega^*)\bar{x}_1^*)$$

from which the conclusion follows.

If  $P_M(j\omega^*, \tau^*) = 0$ , then we can use (22) instead of (20), and follow a similar process.

With Lemma 6, we can easily arrive at the following proposition. **Proposition 2.** Let Assumption IIIa be satisfied, and let  $\theta(\tau)$  and  $\omega(\tau)$  satisfy (C18),  $\theta(\tau^*) = 2r\pi, \omega(\tau^*) = \omega^*$ . Then

$$\mathcal{I}\left(\left(\frac{\partial W}{\partial \omega}\right)_* \overline{\left(\frac{\partial T}{\partial \theta}\right)}_*\right) \neq 0.$$
(C30)

Furthermore,

$$\operatorname{sgn}\left(\mathcal{I}\left(\left(\frac{\partial W}{\partial \omega}\right)_{*}\overline{\left(\frac{\partial T}{\partial \theta}\right)}_{*}\right)\right) = (-1)^{N_{x}(\tau^{*})}\operatorname{sgn}\left(\partial_{\omega}F(\omega^{*},\tau^{*})\right).$$
(C31)

*Proof.* Assume  $P_M(j\omega^*, \tau^*) \neq 0$ . Taking derivative of the equation

$$\Phi(\omega, \tau^*, x_1) = 0,$$

we obtain

$$\frac{\partial \Phi}{\partial \omega} + \frac{\partial \Phi}{\partial x} x_1'(\omega) = 0, \qquad (C32)$$

from which we obtain

$$x_1'(\omega^*) = -\frac{\left(\frac{\partial\Phi}{\partial\omega}\right)_*}{\left(\frac{\partial\Phi}{\partialx}\right)_*} = -\frac{\left(\frac{\partial\Phi}{\partial\omega}\right)_*\left(\frac{\partial\Phi}{\partialx}\right)_*}{\left|\left(\frac{\partial\Phi}{\partialx}\right)_*\right|^2},\tag{C33}$$

where  $(\cdot)_*$  means that the expression is evaluated at  $\omega = \omega^*$ ,  $\tau = \tau^*$ , and  $x_1 = x_1^* = e^{-j\omega^*\tau^*}$ . From the definition of T and W, we obtain

$$\begin{pmatrix} \frac{\partial T}{\partial \theta} \end{pmatrix}_* = \left( \frac{\partial \Phi}{\partial x} \right)_* j x_1^*, \\ \left( \frac{\partial W}{\partial \omega} \right)_* = \left( \frac{\partial \Phi}{\partial \omega} \right)_* + \left( \frac{\partial \Phi}{\partial x} \right)_* (-j \tau^* x_1^*).$$

Therefore,

$$\mathcal{I}\left(\left(\frac{\partial W}{\partial \omega}\right)_* \overline{\left(\frac{\partial T}{\partial \theta}\right)}_*\right) = \mathcal{I}\left(\left(\frac{\partial \Phi}{\partial \omega}\right)_* \overline{\left(\frac{\partial \Phi}{\partial x}\right)_* j x_1^*}\right)$$
$$= -\mathcal{R}\left(\left(\frac{\partial \Phi}{\partial \omega}\right)_* \overline{\left(\frac{\partial \Phi}{\partial x}\right)_* x_1^*}\right)$$
$$= \left|\left(\frac{\partial \Phi}{\partial x}\right)_*\right|^2 \mathcal{R}(x_1'(\omega^*) \overline{x_1^*}).$$

In the last step above, the equation (C33) was used. The conclusion follows from the above equation in view of Lemma 6.

If  $P_M(j\omega^*, \tau^*) = 0$ , then  $P_0(j\omega^*, \tau^*) \neq 0$ . We may use  $\Psi$  instead of  $\Phi$  and follow a similar process.

## Appendix D Proof of Theorem 2

We first prove the following Lemma.

**Lemma 7.** Let Assumption IIIa be satisfied. Let  $\omega(\tau)$ ,  $\theta(\tau)$  and  $\lambda(\tau)$  be defined as in Theorem 2. If

$$\frac{d^k \theta(\tau^*)}{d\tau^k} = 0, \quad k = 1, 2, \dots, l-1,$$
(D34)

then

$$\mathcal{R}\left(\frac{d^k\lambda(\tau^*)}{d\tau^k}\right) = 0,\tag{D35}$$

$$\mathcal{I}\left(\frac{d^k\lambda(\tau^*)}{d\tau^k}\right) = \frac{d^k\omega(\tau^*)}{d\tau^k}, \quad k = 1, 2, \dots, l-1,$$
(D36)

and

$$\mathcal{R}\left(\frac{d^{l}\lambda(\tau^{*})}{d\tau^{l}}\right) = \frac{\mathcal{I}\left(\left(\frac{\partial W}{\partial\omega}\right)_{*}\overline{\left(\frac{\partial T}{\partial\theta}\right)_{*}}\right)}{\left|\left(\frac{\partial D}{\partial\lambda}\right)_{*}\right|^{2}}\left(\frac{d^{l}\theta(\tau^{*})}{d\tau^{l}}\right),\tag{D37}$$

where  $(\cdot)_*$  means that the expression is evaluated at  $\tau = \tau^*$ ,  $\lambda = j\omega^*$ ,  $\omega = \omega^*$  and  $\theta = 2r\pi$ .

Proof. Assumption IIIa and Lemma 3 imply that

$$\partial_{\lambda} D(\lambda, \tau^*)|_{\lambda = j\omega^*} \neq 0.$$
 (D38)

Therefore, the equation

$$D(\sigma + jw, \tau) = 0 \tag{D39}$$

determines continuous functions  $\sigma(\tau) = \mathcal{R}(\lambda(\tau))$  and  $w(\tau) = \mathcal{I}(\lambda(\tau))$  differentiable up to Nth order in a small neighborhood of 0 and  $\omega^*$ , respectively, for  $\tau$  in a small

neighborhood of  $\tau^*$ . Define

$$S(\sigma, w, \tau) = D(\sigma + jw, \tau), \tag{D40}$$

then  $\sigma(\tau)$  and  $w(\tau)$  are determined by

$$S(\sigma, w, \tau) = 0. \tag{D41}$$

Taking the kth order derivative and applying Corollary 3, we obtain

$$\begin{pmatrix} \frac{d^k S}{d\tau^k} \end{pmatrix}_* = \left( \frac{\partial S}{\partial \sigma} \right)_* \left( \frac{d^k \sigma}{d\tau^k} \right)_* + \left( \frac{\partial W}{\partial \omega} \right)_* \left( \frac{d^k w}{d\tau^k} \right)_* + (U_{Sk})_*$$
$$+ V_{Wk} |_{\omega = \omega^*, \tau = \tau^*, \omega^{(i)}(\tau^*) = w^{(i)}(\tau^*), i = 1, 2, \dots, k-1$$
$$= 0,$$
(D42)

where  $U_{Sk} = U_{Sk}(\sigma, w, \tau; \sigma', \sigma'', \dots, \sigma^{(k-1)})$  is a polynomial of  $\sigma'(\tau), \sigma''(\tau), \dots, \sigma^{(k-1)}$  that satisfies

$$U_{Sk}(\sigma, w, \tau; 0, 0, \dots, 0) = 0.$$
 (D43)

In arriving at (D42), we have used the fact that

$$S(\sigma, w, \tau)|_{\sigma=0} = W(\omega, \tau)|_{\omega=w} \,.$$

By the assumption (D34), (C20) is satisfied for k = l. Therefore, (C21) is satisfied for k = 1, 2, ..., l. Apply (D34) and (C21) in (C19), we obtain

$$\left(\frac{\partial W}{\partial \omega}\right)_* \left(\frac{d^k \omega}{d\tau^k}\right)_* + \left(V_{Wk}\right)_* = 0, \quad k = 1, 2, \dots, l-1.$$
(D44)

Using (D43) and (D44), it can be easily verified that

$$\left(\frac{d^k\sigma}{d\tau^k}\right)_* = 0,\tag{D45}$$

$$\left(\frac{d^k w}{d\tau^k}\right)_* = \left(\frac{d^k \omega}{d\tau^k}\right)_*, \ k = 1, 2, \dots, l-1,$$
(D46)

satisfy the equation (D42) for k = 1, 2, ..., l - 1. For k = l, the third term of (D42) still vanish due to (D43). Furthermore, the last term satisfies

$$V_{Wl}|_{\omega=\omega^*,\tau=\tau^*,\omega^{(i)}(\tau^*)=w^{(i)}(\tau^*),i=1,2,\dots,l-1}=(V_{Wl})_*$$

due to (D46). Therefore,

$$\left(\frac{\partial S}{\partial \sigma}\right)_* \left(\frac{d^l \sigma}{d\tau^l}\right)_* + \left(\frac{\partial W}{\partial \omega}\right)_* \left(\frac{d^l w}{d\tau^l}\right)_* + (V_{Wl})_* = 0.$$
(D47)

The above can be solved for  $\left(\frac{d^l\sigma}{d\tau^l}\right)_*$  by multiplying  $\overline{\left(\frac{\partial W}{\partial\omega}\right)}_*$  and taking the imaginary part,

$$\left(\frac{d^{l}\sigma}{d\tau^{l}}\right)_{*} = -\frac{\mathcal{I}\left(\left(\frac{\partial W}{\partial\omega}\right)_{*}(V_{Wl})_{*}\right)}{\mathcal{I}\left(\left(\frac{\partial W}{\partial\omega}\right)_{*}\left(\frac{\partial S}{\partial\sigma}\right)_{*}\right)}.$$
 (D48)

But by definition,

$$\left(\frac{\partial S}{\partial \sigma}\right)_* = \left(\frac{\partial D}{\partial \lambda}\right)_*,\tag{D49}$$

$$\left(\frac{\partial W}{\partial \omega}\right)_* = j \left(\frac{\partial D}{\partial \lambda}\right)_*.$$
 (D50)

Therefore,

$$\left(\frac{d^{l}\sigma}{d\tau^{l}}\right)_{*} = \frac{\mathcal{I}\left(\overline{\left(\frac{\partial W}{\partial\omega}\right)}_{*}(V_{Wl})_{*}\right)}{\left|\left(\frac{\partial D}{\partial\lambda}\right)_{*}\right|^{2}}.$$
(D51)

The equation (D37) can be obtained by comparing (C22) and (D51), and the fact that

$$\mathcal{I}\left(\overline{\left(\frac{\partial W}{\partial \omega}\right)}_{*}\left(\frac{\partial T}{\partial \theta}\right)_{*}\right) = -\mathcal{I}\left(\left(\frac{\partial W}{\partial \omega}\right)_{*}\overline{\left(\frac{\partial T}{\partial \theta}\right)}_{*}\right).$$
 (D52)

With the above lemma, Theorem 2 becomes obvious.

**Proof of Theorem 2.** Apply Lemma 7 for l = m, then the equation (74) is satisfied in view of the equation (D35). To show (75), apply (C31) in Proposition 2 to (D37) in Lemma 7.

## References

- Q. An, E Beretta, Y. Kuang, C. Wang, and H. Wang. Geometric stability switch criteria in delay differential equations with two delays and delay dependent parameters, *Journal of Differential Equations*, 2019, 266(11): 7073-7100.
- [2] S. Barnett. Polynomials and linear control systems, Marcel Dekker, New York, 1983.
- [3] E. Beretta and Y. Kuang. Geometric stability switch criteria in delay differential systems with delay dependent parameters, SIAM Journal on Mathematical Analysis, 2002, 33(5): 1144-1165.
- [4] F. G. Boese. Stability with respect to the delay: On a paper by K.L. Cooke and P. van den Driessche, *Journal of Mathematical Analysis and Applications*, 1998, 228(2): 293-321.

- [5] J. Cai, Q. Gao, Y. Liu, and A. Wu. Generalized Dixon resultant for strongindependent stability of linear systems with multiple delays, *IEEE Transactions* on Automatic Control, 2024, 69(4):2697-2704.
- [6] J. Chen, G. Gu, and Carl N. Nett. A new method for computing delay margins for stability of linear delay systems, Systems & Control Letters, 1995, 26(2): 107-117.
- [7] J.N. Chiasson, S.D. Brierley, and E.B. Lee. Corrections to 'A simplified derivation of the Zeheb-Walach 2-D stability test with application to time-delay systems', *IEEE Transactions on Automatic Control*, 1986, 31(1): 91-92.
- [8] K. L. Cooke and P. van den Driessche. On zeroes of some transcendental equations, *Funkcialaj Ekvacioj*, 1986, 29(1): 77-90.
- [9] F. Crauste. Global asymptotic stability and Hopf bifurcation for a blood cell production model, *Mathematical Biosciences and Engineeringon*, 2006, 3(2): 325-346.
- [10] L. E. El'Sgol'ts and S. B. Norkin. Introduction to the Theory and Application of Differential Equations with Deviating Arguments, Translated by J. L. Casti, Academic Press, New York, 1973.
- [11] H. Fazelinia, R. Sipahi, and N. Olgac. Stability robustness analysis of multiple time-delayed systems using "building block" concept, *IEEE Transactions on Automatic Control*, 2007, 52(5): 799-810.
- [12] K. Gu. A review of some subtleties of practical relevance for time-delay systems of neutral type, *ISRN Applied Mathematics*, 2012, 2012: Article ID 725783, 46 pages.
- [13] K. Gu, V. L. Kharitonov, and J. Chen. Stability of time-delay systems, Boston, MA, 2003.
- [14] K. Gu and M. Naghnaeian. Stability crossing set for systems with three delays, *IEEE Transactions on Automatic Control*, 2011, 56(1): 11-26.
- [15] K. Gu, S.-I Niculescu, and J. Chen. On stability crossing curves for general systems with two delays, *Journal of Mathematical Analysis and Applications*, 2005, 311(1): 231-253.
- [16] J.K. Hale and W. Huang. Global geometry of the stable regions for two delay differential equations, *Journal of Mathematical Analysis and Applications*, 1993, 178(2): 344-362.
- [17] J. K. Hale and S. M. Introduction to Functional Differential Equations, Springer-Verlag, New York, 2013.

- [18] D. Hertz, E.J. Jury and E. Zeheb. Stability independent and dependent of delay for delay differential systems, *Journal of The Franklin Institute*, 1984, 318(3): 143-150.
- [19] D. Hertz, E.J. Jury and E. Zeheb. Simplified analytic stability test for systems with commensurate time delays, *IEE Proceedings D*, 1984, 131(1): 52-56.
- [20] C. Jin, K. Gu, I. Boussaada and S. I. Niculescu. Stability analysis of a more general class of systems with delay-dependent coefficients, *IEEE Transactions on Automatic Control*, 2019, 64(5): 1989-1998.
- [21] C. Jin, K. Gu, S. I. Niculescu and I. Boussaada. Stability analysis of systems with delay-dependent coefficients subject to some particular delay structure, 2018 European Control Conference, 2018, June 12-15, Limassol, Cyprus.
- [22] E.I. Jury. From J.J. Sylvester to Adolf Hurwitz: A historical review, *Stability Theory* (R. Jeltsch and M. Mansour Ed.), Springer, 1996, 121: 53-65.
- [23] H. Kokame, K. Hirata, K. Konishi, and T. Mori. State difference feedback can stabilize uncertain steady state, 2000 American Control Conference, 2000, 2: 1370-1374.
- [24] H. Kokame and T. Mori. Stability preserving transition from derivative feedback to its difference counterpart, *IFAC 15th Triennual World Congress*, Barcelona, Spain, July 2002.
- [25] T. Kato. Perturbation Theory for Linear Operators, Springer-Verlag, Heidelberg, 1995.
- [26] K. Knopp. Theory of Functions, Parts I and II, Translated to English by F. Bagemihl, Dover, Mineola, NY, 1996.
- [27] M. S. Lee and C. S. Hsu. On the τ-decomposition method of stability analysis for retarded dynamical systems, SIAM Journal on Control and Optimization, 1969, 7(2): 242-259.
- [28] C. Méndez-Barrios, S.I. Niculescu, A. Martínez-González, and A. Ramírez. Characterizingsome improperly posed problems in proportional-derivative control, *International Journal of Robust and Nonlinear Control*, 2022, 32(18): 9452-9474.
- [29] C. Méndez-Barrios, J.-D. Torres-Garcia, and S.I. Niculescu. Delay-difference approximations of PD-controllers: improperly-posed systems with multiple delays case, *International Journal of Robust and Nonlinear Control*, 2024, DOI: 10.1002/rnc.7217.
- [30] W. Michiels and S. I. Niculescu. Stability, control, and computation for time-delay systems: an eigenvalue-based approach, SIAM, 2014.

- [31] S. I. Niculescu. Delay Effects on Stability: A Robust Control Approach, Springer, Heidelberg, 2001.
- [32] R. Nussbaum. Differential delay equations with two time lags, Memoirs of the American Mathematical Society, 1978, 16(205).
- [33] L. Orlando. Sul problema di Hurwitz relativo alle parti reali delle radici di un'equatione algebrica, *Mathematische Annalen*, 1911, 71: 233-245.
- [34] S. Ruan and J. Wei. On the zeros of transcendental functions with applications to stability of delay differential equations with two delays, *Dynamics of Continuous Discrete and Impulsive Systems Series A*, 2003, 10: 863-874.
- [35] R. Sipahi and N. Olgac. Stability analysis of multiple time delay systems using the direct method, ASME International Mechanical Engineering Congress and Exposition, Washington, DC, November 2003.
- [36] I. H. Suh and Z. Bien. Proportional minus delay controller, *IEEE Transactions on Automatic Control*, 1979, 24(2): 370-372.
- [37] I. H. Suh and Z. Bien. Use of time-delay actions in the control design, IEEE Transactions on Automatic Control, 1980, 25(3): 600-603.
- [38] D. Torres-Garcia, C.-F. Méndez-Barrios, S.-I. Niculescu, and A. Martínez-González. Delay-difference approximation of PD-controllers. Insights into improperly-posed closed-loop systems, 1st IFAC Workshop on Control of Complex Systems, Bologna, Italy, 2022, 55(40): 79-84.
- [39] K. Walton and J. E. Marshall. Direct method for TDS stability analysis, *IEE Proceedings D*, 1987, 2(134): 101-107.
- [40] A. L. Wilmot-Smith, D. Nandy, G. Hornig, and P.C.H. Martens. A time delay model for solar and stellar dynamos, *The Astrophysical Journal*, 2006, 652: 696-708.
- [41] N.J. Young. An identity which implies Cohn's theorem on the zeros of a polynomial, Journal of Mathematical Analysis and Applications, 1979, 70(1): 240-248.
- [42] C. Yuan, S. Song, Q. Gao, H.R. Karimi, L. Pekar, and S. Guo. A novel frequency domain approach for the exact range of imaginary spectra and the stability analysis of LTI systems with two delays, *IEEE Access*, 2020, 8:36595-36601.